Author's response to reviews

Title: Knowledge of vitamin D and perceptions and attitudes towards sunlight among Chinese middle-aged and elderly women: a population survey in Hong Kong

Authors:

Annie WC Kung (awckung@hkcc.hku.hk)
Kar-Kui Lee (kklee@graduate.hku.hk)

Version: 4 Date: 4 August 2006

Author's response to reviews:

Dr Alexander Pemberton
Editor
BMC Public Health
22.06.2006

Dear Dr Pemberton

Re: Manuscript ID 2143553974104150
Knowledge of Vitamin D and perceptions and attitudes towards sunlight among Chinese Middle-aged and elderly women: a population survey in Hong Kong

Thank you for your letter together with the reviewers' comments. We thank the reviewers for their helpful comments and we have amended the manuscript accordingly.

Page 11, Discussion 2nd last sentence is changed to read "SPF>15 was chosen because sunscreen products with SPF<15 are not commercially available in Hong Kong."

Attached please find the amended manuscript. I hope you would reconsider publishing our manuscript.

Yours Sincerely,

Annie Kung